

The influence of qualified foreign institutional investors on internal control quality: evidence from China

Article

Accepted Version

Creative Commons: Attribution-Noncommercial-No Derivative Works 4.0

Li, Z., Wang, B., Wu, T. and Zhou, D. ORCID: https://orcid.org/0000-0003-4238-0526 (2021) The influence of qualified foreign institutional investors on internal control quality: evidence from China. International Review of Financial Analysis, 78. 101916. ISSN 1057-5219 doi: 10.1016/j.irfa.2021.101916 Available at https://centaur.reading.ac.uk/100493/

It is advisable to refer to the publisher's version if you intend to cite from the work. See <u>Guidance on citing</u>.

To link to this article DOI: http://dx.doi.org/10.1016/j.irfa.2021.101916

Publisher: Elsevier

All outputs in CentAUR are protected by Intellectual Property Rights law, including copyright law. Copyright and IPR is retained by the creators or other copyright holders. Terms and conditions for use of this material are defined in the <u>End User Agreement</u>.

www.reading.ac.uk/centaur



CentAUR

Central Archive at the University of Reading

Reading's research outputs online

The Influence of Qualified Foreign Institutional Investors on Internal

Control Quality: Evidence from China

Zhe Li^a, Bo Wang^b, Tianlong Wu^c, Dan Zhou^{d,*}

^a Department of Accounting and Finance, School of Management, Swansea University, UK SA1 8EN; Email: zhe.li@swansea.ac.uk; ORCID: <u>https://orcid.org/0000-0003-1442-4499</u>.

^b Department of Banking and Finance, Southampton Business School, Building 2, Highfield Campus, SO17 1BJ; Email: <u>B.Wang@soton.ac.uk</u>; ORCID: <u>https://orcid.org/0000-0001-9417-2214</u>.

^c Beixin Ruifeng Fund Management, Haidian District, Beijing, P.R. China; Postcode: 110108; Email: <u>wtl1993@yeah.net</u>.

^{d.}**Corresponding author*, Business Informatics Systems & Accounting, Henley Business School, University of Reading, Whiteknights campus, UK RG6 6UD; Email: <u>dan.zhou@henley.ac.uk</u>; ORCID: <u>http://orcid.org/0000-0003-4238-0526</u>.

Forthcoming, International Review of Financial Analysis

Accepted version

ACKNOWLEDGEMENT: We would like to thank Professor Brian Lucey (the Editor) and the anonymous referee for their constructive comments. All remaining errors are our own.

Abstract

This study aims to investigate whether qualified foreign institutional investors (QFIIs) improve companies' internal control quality to mitigate information asymmetry. By analysing a sample of 22,310 firm-year observations from Chinese listed companies between 2005 and 2017, we found that companies with QFIIs exhibit higher internal control quality and fewer internal control deficiencies. In particular, higher QFII ownership is associated with higher quality of internal control. Interestingly, QFIIs from high-governance-quality countries are more likely to improve the internal control system of their investee companies. Finally, the improvement in internal control quality attributed to QFIIs leads to better operating performance. Our results are robust to alternative measures of QFIIs, alternative proxies for internal control quality, and various controls for endogeneity issues.

Keywords: Qualified foreign institutional investors; internal control quality; Chinese equity markets; information asymmetry

JEL classification: G23, G30, L25

1. Introduction

Foreign institutional investors play an increasingly important role in the integration of international capital markets (Jin et al., 2016). They are one of the leading players in the globalisation of emerging markets. Nevertheless, as minority and "outside" investors, foreign institutional investors usually suffer from severe information asymmetry that is caused by multiple agency issues (principal-principal and principal-agent) and/or an inferior information environment within investee companies (Kang and Kim, 2010; Huang and Zhu, 2015; Zhang et al., 2017; Kim et al., 2020). Therefore, overcoming the information disadvantage is crucial for foreign institutional investors (Kim et al., 2020). However, we still lack knowledge on how foreign institutional investors mitigate information asymmetry. Further, investigating the influence of foreign institutional investors on specific governance practices and corporate events in emerging markets is also an important but unexplored area (Aggarwal et al., 2011; Huang and Zhu, 2015). Relevant studies mainly focus on auditor choices (He et al., 2014; Kim et al., 2019), global convergence of accounting reports (Fang et al., 2015), split-share structure reform (Huang and Zhu, 2015), and earnings management (Lel, 2019). Our study seeks to address this gap by investigating the influence of foreign institutional investors on the effectiveness (high-quality) of internal control, which is a vital governance practice to address agency problems and information asymmetry.

Internal control refers to a process designed to provide reasonable assurance regarding the achievement of objectives in the following categories: effectiveness and efficiency of operations, reliability of financial statements, and compliance with applicable laws and regulations (Changchit et al., 2001; Doyle et al., 2007). An effective internal control system has long been considered as a crucial corporate governance mechanism to prevent the recurrence of unethical conduct (e.g. earnings management and fraud) and reporting improprieties (Ji et al., 2017). In particular, a high-quality internal control system ensures high-

1

quality financial and non-financial information reporting and a transparent information environment (Ashbaugh-Skaife et al., 2007; Feng et al., 2009; Ji et al., 2017), and hence, this mitigates information asymmetry. This is particularly critical for minority investors, such as foreign institutional investors, because their direct influence on business strategies is limited. A well-established internal control system can complement weak monitoring roles and reduce potential investment risks (Chen et al., 2017).

The Chinese setting provides an ideal laboratory to study whether foreign institutional investors will improve investee companies' internal control quality for the following two reasons. First, since the Qualified Foreign Institutional Investor (QFII) scheme was officially launched by the China Securities Regulatory Commission (CSRC) in 2002, we have observed an increasingly large number of foreign institutional investors entering Chinese markets, especially those from well-governed jurisdictions. However, because of the quota system, QFIIs can only take small ownership and face agency problems from management and large domestic investors, such as state entities. Hence, QFIIs are more likely to protect their benefits by improving corporate governance practices to reduce agency issues (Huang and Zhu, 2015; Cao et al., 2017). This, in turn, enhances their local reputation in China and leads to more investment quotas. Meanwhile, these qualified offshore investors are experienced and sophisticated large international investors who have advanced knowledge and skills to monitor and advise business and governance practices (Kim et al., 2020; Li et al., 2021). The QFII scheme expects them to not only facilitate the Chinese market's financial globalisation but also advise better and internationally accepted governance practices (Huang and Zhu, 2015). Although China has achieved significant economic progress and has become the largest emerging economy in the world, its inferior governance, weak institutions, and poor law enforcement still raise severe concerns (Bai et al., 2004; Yuan et al., 2009).

Based on a panel dataset of 2,773 unique Chinese listed companies with 22,310 firmyear observations between 2005 and 2017, we find that companies with QFIIs experience an increase in future internal control quality and exhibit fewer internal control deficiencies. Similarly, QFII ownership is positively associated with future internal control quality. We then explore why QFIIs push investee companies to improve the quality of internal control. First, the incentive to facilitate the effectiveness of internal control may be attributable to QFIIs' high awareness of corporate governance because the overwhelming majority of these investors in Chinese listed companies are from well-governed regimes or advanced economies, such as countries in North America and Western and Northern Europe. As a result, they are more accustomed to higher-standard codes of conduct and better governance practices in their home countries (Gong et al., 2013).¹ Hence, when investing in foreign markets, these offshore investors are highly likely to transplant their strong governance awareness and high standards of conduct to the investees, more likely to comply with rules and laws, and pay particular attention to internal control issues. Second, investing in a foreign market is accompanied by additional risk and investment uncertainty due to a lack of transparent and sufficient information for the fair evaluation of their prospective investees; when compared to local investors, overseas investors are naturally characterised by information disadvantages (Oh et al., 2011; Li et al., 2021). Because efficient internal control is generally viewed as an essential signalling mechanism that helps reduce information asymmetry and enhance transparency (Feng et al., 2009; Ji et al., 2017), foreign investors are motivated to promote better internal control practices and help maintain the effectiveness of the control system to address concerns that arise from geographical distances. Our evidence reveals that this positive impact is more salient in companies with QFIIs from countries with high institutional quality than in

¹ Similarly, Jia et al. (2020) report that 95.83% of QFIIs in Chinese companies come from advanced economies deemed by the IMF.

companies with QFIIs from countries with low institutional quality. Our findings are robust to a series of additional tests, including alternative measures of key variables, controlling for the characteristics of the audit committee and the effects of domestic institutional investors, as well as a subsample excluding manufacturing entities. Our results are consistent after conducting the dynamic panel generalised method of moments (GMM) and propensity score matching (PSM) approach.

We further uncovered the mechanism for the positive influence of QFII-licensed investors. Existing literature has documented that key shareholders can influence corporate policies by proposing and voting on the board (Lee and Lounsbury, 2011; Oh et al., 2011; Li et al., 2021). Thus, we identify whether QFII is ranked among the top ten shareholders of the company to capture the strength of the influence of QFIIs. Our results demonstrate that foreign investors, when they are among the top ten largest shareholders, have a greater influence on decision making and offer a key mechanism that promotes toward effective internal control. Finally, our further analysis reveals that the enhanced internal control quality attributed to QFIIs leads to higher financial performance.

This study contributes significantly to the literature in three ways. First, we provide new evidence for the ever-growing number of studies exploring the motives and roles of foreign institutional investors. Existing literature on foreign investors' impact mainly focuses on the integration of technological, human, and financial resources (Huang and Shiu, 2009), knowledge spillovers and innovation (Luong et al., 2017), financial stability (Schuppli and Bohl, 2010), stock price crash risk (Kim et al., 2020), social responsibility (Dyck et al., 2019), and firm performance (Douma et al., 2006). This study underscores the effect of QFIIs on improving the quality of firm-level internal control and reducing internal control deficiencies in China. Second, we echo the call for investigating the influence of foreign institutional investors on specific corporate governance practices (Ferreira and Matos, 2008; Aggarwal et al., 2011; Huang and Zhu, 2015). The current literature mainly focuses on the influence on governance index, which leads to less guidance on practices and regulations. Importantly, we empirically demonstrate that this impact is driven by QFIIs from countries with high institutional quality.

Third, our research provides new insights into current literature on the determinants of internal control quality. Existing literature has documented that the quality of internal control is influenced by firm-specific characteristics (Chen and Keung, 2018), board heterogeneity (Hu et al., 2017), ownership structure (e.g., ownership concentration, managerial, and family ownership) (Deumes and Knechel, 2008; Bardhan et al., 2015), characteristics of the audit committee (Zhang et al., 2007), senior executives' functional backgrounds (Yu et al., 2019; Oradi et al., 2020), national culture, and market regulations.² Our study highlights the importance of foreign investors, especially their higher awareness of governance and strict standards of conduct, as a critical channel to facilitate the effectiveness of the internal control system for companies in countries with inferior corporate governance and weak institutional environments.

The remainder of this paper is organised as follows. Section 2 reviews the current literature and develops our hypotheses. The research design is presented in Section 3. Section 4 presents the main findings, robustness checks, and endogeneity analyses. Section 5 explores the mechanism by which QFIIs may be used to promote for changes in the internal control quality of a company. Section 6 extends the study, and Section 7 concludes the paper.

² See Chalmers et al. (2019) for a detailed review.

2. Literature review and hypotheses development

2.1 Institutional background

In November 2002, China partially opened its domestic financial market to foreign institutional investors by launching a scheme assigning investment quotas to QFIIs, which was officially approved by the CSRC. This scheme aims to develop domestic capital markets gradually and allows QFII-licensed entities to buy and sell Chinese Yuan (CNY)-denominated A-shares³ listed on the Shenzhen Stock Exchange (SZSE) and Shanghai Stock Exchange (SSE). Since then, international investors in the domestic A-share market have dramatically increased with respect to foreign investment quotas and the number of QFIIs. The motivation for implementing the QFII scheme is as follows: First, because of the strict selection criteria of the scheme, QFIIs are large and internationally famous institutions (Huang and Zhu, 2015) with in-depth investment knowledge and a strong sense of governance awareness (Gul et al., 2010; Huang and Zhu, 2015; Jin et al., 2016). Therefore, the Chinese market, as a relative latecomer to international markets, attempts to rely on QFIIs to effectively integrate human capital and facilitate asset allocation, in addition to introducing superior managerial skills and corporate governance practices. Second, the QFII scheme aims to help enhance the monetary system, gradually internationalise, and stabilise the Chinese capital market (Han et al., 2015). Third, QFII-licensed entities are expected to suppress the impact of overseas speculative "hot money" on the domestic economy, building up an open, competitive, and orderly modern market system (Huang and Zhu, 2015).

Because corporate ownership in China is highly concentrated and a well-developed internal control framework does not exist, controlling shareholders can easily reap private benefits of control by extracting value from the company to the detriment of minority

³ A-shares refer to companies that are incorporated in China and traded on the Shanghai and Shenzhen exchanges; they are quoted in local renminbi and entail foreign investment regulations.

shareholders. Hence, foreign investors may face severe principal–principal agency problems and information asymmetry (He et al., 2014). In addition, internal control deficiencies and weak internal audits (manifested in less reliable financial reporting) could raise severe information asymmetry between outside investors and insiders, thus increasing firm risk (Ashbaugh-Skaife et al., 2009). Extensive studies have acknowledged that an effective internal control system can provide high-quality assurance, regarding the efficiency of an entity's business operational environment and risk management, reliability of financial reporting and transparency of information disclosure, achievement of the monitoring of internal control deficiencies, and compliance with applicable laws, policies, and regulations (Changchit et al., 2001; Doyle et al., 2007; Chen et al., 2017), thereby serving as a potential mechanism for mitigating conditions of information asymmetry and managerial opportunistic behaviours. Therefore, foreign investors may have incentives to alleviate their information disadvantages and reduce investment risk by enhancing the internal control of investee companies.

In response to the increasing demand for enhanced governance in addressing internal control deficiencies, internal audit, and financial restatement issues (Hu et al., 2017), Chinese regulators have promoted internal control practices and helped facilitate an effective internal control system among listed firms. Indeed, China has made substantial progress in facilitating the effectiveness of internal control over the past few years by introducing a series of localised guidelines. For instance, in May 2008, the *Basic Standard of Enterprise Internal Control* was jointly issued by five regulatory authorities.⁴ Later in December 2008, the SSE and SZSE issued the *Notice on the Preparation of 2008 Annual Financial Reports by Public Firms*, a notice which requires listed firms to disclose annual self-evaluation reports on internal control.⁵

⁴ See Ji et al. (2017) and <u>http://www.mof.gov.cn/gkml/caizhengwengao/</u>.

⁵ Guidelines for Evaluation of Enterprise Internal Controls and Guidelines for Auditing of Enterprise Internal Controls can be found at <u>http://www.gov.cn/zwgk/2010-05/05/content_1599512.htm</u>.

These efforts suggest that internal controls should be in accordance with higher international standards. Therefore, we believe that the current study is important because the opening of the domestic market to foreign institutional investors, especially those from well-governed countries, is highly likely to have a significant influence on corporate governance, motivating domestic companies to further facilitate the construction of efficient internal control systems.

2.2 Hypothesis development

As mentioned above, QFIIs tend to suffer from severe information disadvantages and higher levels of investment environment uncertainty and risk, due to cultural differences, geographical distances, and minority ownership (Liu et al., 2014; Li et al., 2021). Therefore, they usually have strong incentives and resources/skills/expertise at their disposal to protect their investments in their portfolio firms. Extant literature has documented the influence of QFIIs on Chinese listed companies to mitigate agency problems and information asymmetry. For instance, Huang and Zhu (2015) find that QFII-licensed investors help float non-tradable shares and increase compensation for minority tradable shareholders. Foreign investing parties can reduce the likelihood of earnings management (Lel, 2019). Kim et al. (2020) also find that QFII-licensed investors help reduce Chinese listed companies' stock price crash risk through their external monitoring. Li et al. (2021) argue that QFIIs from high-regulatory-quality jurisdictions tend to introduce their social awareness to investee companies.

An effective internal control system will facilitate the transparency of the information environment, management forecast accuracy, and information quality (Ashbaugh-Skaife et al., 2007; Feng et al., 2009; Ji et al., 2017), which will reduce investors' investment risks. Moreover, an effective internal control system will ultimately drive operational efficiency and financial performance. For example, the disclosure of internal control weaknesses and improvement in internal control quality significantly mitigates firm-level investment inefficiency (Cheng et al., 2013). An effective internal control system serves as a crucial mechanism to mitigate the loss of economic value and offset risks in negative events (Wang et al., 2018). QFIIs can share the financial success of the entities in which they invest. As such, one would expect that QFIIs also have strong incentives to pay particular attention to internal control issues and pressure management to improve the quality and effectiveness of the internal control system, apart from other corporate governance mechanisms.

In addition, QFIIs are large internationally well-known investors who are experienced in internationally accepted advanced corporate governance practices. This is because corporate governance, including internal control practices, has been well-developed and adopted for several decades in well-governed economies such as the UK and North America (Maijoor, 2000), which are QFIIs' home countries or main investment markets. When investing in the Chinese market, which is characterised by weakly governed institutional environments and inferior internal control, QFIIs are expected by the government and companies to advise good corporate policies (Jiang and Kim, 2015) including internal control practices. They will also transplant their strong awareness of corporate governance and high standards of conduct to the companies in which they invest (Li et al., 2021). This, in turn, will help QFIIs improve their reputations in the Chinese market and may facilitate an increase in their investment quota. Therefore:

Hypothesis 1: Firms with QFIIs exhibit better internal control quality than those without QFIIs, ceteris paribus.

When investigating the positive effect of QFIIs on internal control quality, it is necessary to review their home countries' institutional quality that may influence their distinctive governance behaviours and awareness. National governance backgrounds can largely explain the disparities in governance practices across countries (La Porta et al., 2008; Del Bosco and Misani, 2016; Li et al., 2021). From an institutional perspective, a high level of country-level regulation places pressure on companies to comply with institutional guidelines, requirements, and rules, thereby improving the effectiveness and quality of internal control (DiMaggio and Powell, 1983; Chalmers et al., 2019; Li et al., 2021). Further, Del Bosco and Misani (2016) provide evidence that companies headquartered in countries with higher institutional quality, measured as high Worldwide Governance Indicator (WGI) scores, tend to have better governance systems which ensure their board members and executives act in the best interests of its shareholders, thereby confirming that the quality of institutions is a key driving force behind corporate governance. According to the WGI, institutions in countries with higher national governance quality are more likely to abide by the rules and laws of society and have better enforcement of contracts, investor protection, transparency, and accountability of the governance system and integrity, thereby enhancing the stringency of the business regulatory environment (Del Bosco and Misani, 2016; Li et al., 2021).⁶ These characteristics are highly associated with the construction of an effective internal control system.

Well-governed countries largely depend on regulation-based mechanisms that place restrictions on ex-ante behaviours (La Porta et al., 2008). Therefore, QFIIs from highinstitutional-quality countries are more likely to exhibit a higher awareness of corporate governance and stricter standards of codes of conduct and comply with the rules and laws to construct governance systems. Hence, when they invest in foreign markets, they may carry out their strong governance behaviours by facilitating internal monitoring and the construction of efficient internal control systems among the companies in which they invest, thereby improving their internal control quality.

Hypothesis 2: The positive influence of QFIIs on internal control quality is more pronounced in companies with QFIIs from high-institutional-quality countries than in companies with QFIIs from low-institutional-quality countries, ceteris paribus.

⁶ Details are available at <u>https://info.worldbank.org/governance/wgi/Home/Documents</u>.

3. Research design

3.1 Sample and collection

Our firm-level sample includes all Chinese A-share companies listed on either the SSE or the SZSE with internal control index scores from 2005 to 2017. The scores are obtained from the Dibo Internal Control and Risk Management Database,⁷ which evaluates the internal control activities of Chinese listed companies by assessing public information from the CSRC, SSE, SZSE, domestic and foreign media, and regulatory authorities. This database is widely used and cited by media, auditors, corporations, and scholars in China (Chen et al., 2017; Wang et al., 2018; Li et al., 2019).

We extract the data on QFII identities and ownership characteristics, such as foreign institutional ownership, name, and headquarters of each foreign institution, from the Wind-Financial Platform and the State Administration of Foreign Exchange.⁸ All financial and governance data were obtained from the China Stock Market and Accounting Research (CSMAR) platform. Companies in the financial industry were excluded from this sample. The final sample consists of 2,773 unique companies with 22,310 firm-year observations with QFIIs from 23 countries.

3.2 Key variables

The internal control quality (ICI) is measured by dividing the Dibo internal control index (score) by 10, which alleviates the concern of a relatively scattered distribution of the index (Wang et al., 2018). The Dibo internal control index (score) is a measure of the internal control efficiency at the aggregate level of a company in a given fiscal year. Specifically, the evaluation process involves firm and industry risk, internal audit, internal control deficiencies,

⁷ Dibo Internal Control and Risk Management Database was developed by China Shenzhen DIB Company and supported by Sun Yat-sen University and the China Ministry of Finance. It is a leading internal control information provider in China. The Dibo internal control scores are available at http://www.ic-erm.com/index.html. Our sample ranges from 2005 to 2017, commencing in 2005 because the number of firms with QFIIs before then were too few to enable meaningful comparison. ⁸ See https://www.wind.com.cn/ and https://www.safe.gov.cn/guangdong/2019/0107/1293.html.

violations of laws and regulations, related-party transactions, litigation, laws, and regulations. This score is constructed based on five aspects: internal control environment, risk assessment, internal monitoring, control activities, and information and communication, and ranges from 1 to 1000, with a higher value of the index representing higher internal control quality.

We also employ two alternative measurements to proxy for internal control quality. First, since the internal control score varies widely across industries, we utilise an industrymedian-adjusted score, *ADJ_ICI*, which is measured as the deduction of the internal control index of a company from the median index for peers in the same industry during the year.⁹ Second, we use the number of internal control deficiencies (*NO_DEFI*) because the disclosure of internal control weaknesses and deficiency information sends a key message to evaluate a firm's internal control system (Hu et al., 2017; Oradi et al., 2020). Fewer internal control deficiencies within a company imply an efficient internal control system.

Following Huang and Zhu (2015) and Li et al. (2019), we use a categorical variable (*QFII_DUMMY*) to measure the presence of QFIIs. *QFII_DUMMY* equals one if a company has at least one QFII in a fiscal year and zero otherwise. We also create a continuous variable, *QFII_OWN*, measured as the percentage of outstanding shares held by QFIIs, to capture the magnitude of the effect of foreign ownership.

3.3 Model specification

To test our hypotheses, we specify the following ordinary least squares (OLS) regression with year and industry fixed effects:

 $ICI_{i,t} = \alpha + \beta_1 QFII_DUMMY_{i,t-1} + \beta_2 Control_{i,t-1} + Year FE + Industry FE + \varepsilon_{i,t}$ (1) where *ICI* represents the internal control score, which can be replaced by the industry-medianadjusted internal control score (*ADJ_ICI*) and the number of internal control deficiencies (*NO_DEFI*). *QFII_DUMMY* denotes the presence of QFII-licensed investors and can be

⁹ The industry classification follows the 2012 CSRC industry categories.

replaced by *QFII_OWN*. We expect β_1 to be significantly positive (negative) if the dependent variable is *ICI* or *ADJ_ICI* (*NO_DEFI*).

We controlled for a wide range of factors (*control*). We account for firm size (*FIRMSIZE*) and firm age (*FIRMAGE*) because prior literature documents that larger and older firms are likely to exhibit better internal control quality (Chen and Keung, 2018). We also include leverage (*LEVERAGE*) because high-levered firms may have better internal control under lenders' monitoring (Wu and Yue, 2009). High-growth firms are likely to have internal control weaknesses (Hu et al., 2017); therefore, we control for *GROWTH*. To the extent that internal control quality might differ for firms audited by large audit firms, we include an indicator variable, *BIG4*, to indicate companies audited by the Big Four accounting organisations. The quality of an entity's internal control is a function of the effectiveness of the board of directors (Hu et al., 2017; Oradi et al., 2020). For instance, board size and independence are found to be negatively related to the likelihood of financial fraud and regulatory authority enforcement actions. Hence, we control for board size (*BOARDSIZE*), board independence (*INDEP*), CEO duality (*DUALITY*), and board meetings (*MEETING*). We also control for financial health (*LOSS*), state ownership (*SOE*), and ownership concentration (*OWN_CON*) (Oradi et al., 2020). The definitions of all variables can be found in Appendix A.

In all regressions, standard errors are corrected for heteroskedasticity and clustered at the firm and year levels (Petersen, 2009; Thompson, 2011). We winsorize all continuous variables at the 1st and 99th percentiles of their respective distributions.

4. Empirical results

4.1 Descriptive statistics

Table 1 presents the sample distribution across years (Panel A) and industries (Panel B). Panel A indicates that about 8.8% (1,956/2,2310) of the observations are backed by QFIIs. Specifically, the percentage of companies with QFIIs increased to about 10% (268/2,683) in

2017, up from 2.45% (13/530) in 2005, implying that the QFII scheme launched by the Chinese government has significantly facilitated foreign institutional investment in the domestic capital market. ¹⁰ Panel B shows that 58.3% (13,005/22,310) of our observations are from the manufacturing sector. In particular, among 13,005 firm-year observations in the manufacturing sector, 1,198 observations (9.21%) are backed by QFIIs, which is consistent with the findings of Liu et al. (2014) and Li et al. (2021).

[Insert Table 1 here]

Table 2 provides descriptive statistics. *ICI* varies considerably from 0.0000 to 99.5360, with a mean (median) value of 64.7354 (67.6815), which is comparable to the findings of Lu and Cao (2018). This indicates that more than half of the companies in the sample have internal control index scores higher than the average level. The mean (median) value of the industry-median-adjusted internal control index (*ADJ_ICI*) is -2.4127 (0.0000), which varies from - 78.3650 to 68.1430. The mean (median) value of the natural logarithm of the number of internal control deficiencies (*NO_DEFI*) is 0.8299 (0.0000). A total of 20.39% of the sample companies conducted financial restatements. Of the companies, 8.77% have at least one QFII during the sample period, and 4.35% (4.42%) of the sample companies have QFIIs from high-institutional-quality (relatively low-institutional-quality) countries. Of the sample companies, 7.02% have at least one QFII among the top ten largest shareholders (*QFII_TOPTEN*).

[Insert Table 2 here]

Table 3 presents the correlations between the main variables. The correlation coefficient between *QFII_DUMMY* (*QFII_HIGHWGI*) and *ICI* is significantly positive,

¹⁰ Furthermore, we report the distribution by the institutional quality of QFIIs' countries of domicile in Appendix B. Among 1,956 firm-year observations with the presence of QFIIs, 49.59% (970 out of 1,956) of the observations have QFIIs from countries with high institutional quality, and 50.41% (986 out of 1,956) of the observations have QFIIs from countries with low institutional quality.

providing initial support for H1. The low correlations among the explanatory variables suggest that multicollinearity is not a concern in our dataset.

[Insert Table 3 here]

4.2 Main regression results

4.2.1 Effects of QFIIs

Table 4 presents the main results for H1. The coefficient of *QFII_DUMMY* in Model 1 is positive and statistically significant (coefficient=0.5105; t=1.9780), indicating that companies with the presence of QFII-licensed investors exhibit a higher subsequent internal control score than those without. Similarly, in Model 2, the estimate of *QFII_OWN* is positive and highly significant (coefficient=0.2569; t=3.4461) at the 1% level, suggesting that foreign ownership is positively associated with internal control quality. Therefore, H1 is supported.

[Insert Table 4 here]

The results are robust when using the industry-median-adjusted internal control index (*ADJ_ICI*) (Models 3-4) and the number of internal control deficiencies (*NO_DEFI*) (Models 5-6) as the dependent variable. For instance, the estimates for *QFII_DUMMY* and *QFII_OWN* in Models 3-4 are both positive and highly significant, confirming a positive link between QFIIs and future internal control quality. The coefficients of *QFII_DUMMY* and *QFII_OWN* in Models 5-6 are significantly negative, suggesting that companies with QFIIs exhibit fewer internal control deficiencies.

The effects of the control variables are broadly consistent with prior findings in existing literature. Specifically, firm size (*FIRMSIZE*) and ownership concentration (*OWN_CON*) are positively associated with internal control quality, while firm age (*FIRMAGE*), leverage ratio (*LEVERAGE*), sales growth (*GROWTH*), and the indicator of negative income in the prior year (*LOSS*) are negatively related to internal control quality.

4.2.2 Effects of the institutional quality of QFIIs' home countries

To test the validity of H2, we divide QFIIs into two categories according to the institutional quality of QFIIs' home countries. Specifically, following Del Bosco and Misani (2016), Liu et al. (2019), and Li et al. (2021), we employ the WGI scores of the World Bank as a proxy for institutional quality and calculate the median value for each year.¹¹ We then generate two categorical variables. *QFII_HIGHWGI* is assigned a value of one for companies having a QFII-licensed investor from a country/region with an institutional quality score equal to or greater than the median value in a fiscal year, and zero otherwise. *QFII_LOWWGI* is set to one for companies having a QFII from a country/region with an institutional quality score lower than the median value, and zero otherwise. In the case of multiple QFII-licensed investment entities within a company during the year, we follow Li et al. (2021) to identify the dominant QFII institution based on the total percentage of shares by QFIIs.¹² Next, we replace *QFII_DUMMY* with *QFII_HIGHWGI* and *QFII_LOWWGI* in Equation (1) and display the estimates in Models 1, 3, and 5 of Table 5. The estimate of *QFII_HIGHWGI* (*QFII_LOWWGI*) is significant at the 1% level (insignificant) in Models 1 and 3, and the estimate of *QFII_HIGHWGI* (*QFII_LOWWGI*) in Model 5 is significantly negative

¹¹ WGI includes six dimensions: voice and accountability, political stability and absence of violence, government effectiveness, regulatory quality, rule of law, and control of corruption. Consistent with Del Bosco and Misani (2016), we averaged the six indicators (using equal weights) to build a WGI index as a comprehensive national governance quality measure. WGI ranges from -2.5 to 2.5, with higher WGI corresponding to higher levels of institutional quality. Details are available at https://info.worldbank.org/governance/wgi/Home/Documents.

¹² Taking Jiangling Motors Corporation (stock code: 000550, Shenzhen Stock Exchange), one of the largest exporters of light diesel commercial vehicles in China, as an example, Jiangling has three QFII-licensed investment entities in 2014, namely, China International Capital Corporation Hong Kong Asset Management Limited, Canada Pension Plan Investment Board, and Kuwait Investment Authority; these three investors own 1.25%, 1.53%, and 1.03% of the shares issued by the firm, respectively. The WGI scores of Hong Kong (1.51) and Canada (1.65) are higher than the median WGI (1.46), thus a high institutional quality system, while the WGI score of Kuwait (-0.17) is lower than the median value of WGI, thereby a relatively inferior governance system. Hence, the dominant QFII in Jiangling in 2014 is identified as the high-institutional-quality system because 2.78% is greater than 1.03%.

(insignificant), suggesting that the positive influence of QFIIs on internal control quality is mainly attributable to QFIIs from countries/regions with high institutional quality.

[Insert Table 5 here]

Moreover, we introduce two continuous variables to investigate the influence of QFIIs from high- and low-institutional-quality countries/regions on investees' internal control quality. Specifically, *QFII_HIGHWGI_OWN* is measured as the sum of the percentage of outstanding shares held by QFII institutions originating from regions with high institutional quality. *QFII_LOWWGI_OWN* is defined as the sum of the percentage of outstanding shares held by QFIIs from jurisdictions with relatively low institutional quality. We then substitute *QFII_HIGHWGI_OWN* and *QFII_LOWWGI_OWN* for *QFII_DUMMY* in Equation (1) and report the results in Models 2, 4, and 6. Similarly, the estimate of *QFII_HIGHWGI_OWN* is positive (negative) and significant in Models 2 and 4 (6), while that of *QFII_LOWWGI_OWN* is insignificant.

Collectively, our results consistently reveal that the institutional quality of QFIIs can be viewed as an important channel that transplants QFIIs' governance standards to investee companies, and QFIIs with high governance awareness and high-standard codes of conduct promote the effectiveness of internal control. This evidence strongly supports H2.

4.3 Robustness checks

4.3.1 Alternative measures

We employed a few alternative measures to test robustness. First, financial restatements may be symptomatic of poor internal control (Ashbaugh-Skaife et al., 2007; Hu et al., 2017). As such, we follow Hu et al. (2017) and use the information of financial restatements, such as the probability of a financial restatement (*RESTATEMENT*) and the number of financial restatements (*NO_RESTATEMENT*) as alternative proxies for internal control quality. In doing so, we substitute *RESTATEMENT* and *NO_RESTATEMENT*, respectively, for *ICI* in Equation

(1) and report the results of Models 1-2 in Table 6. The estimate of *QFII_DUMMY* in both models is negative and statistically significant, indicating that companies with QFIIs are associated with a lower probability of future restatements and exhibit fewer financial restatements. Thus, the evidence supports H1.

[Insert Table 6 here]

Second, according to Jin et al. (2016), we also employ the number of QFIIs (NO_QFII) as an alternative measure to capture the influence of QFIIs and re-estimate Equation (1). The results in Model 3 of Table 6 show a positive and significant coefficient (t=4.2243) on NO_QFII , reconfirming our main findings again.

4.3.2 Accounting for characteristics of the audit committee

The audit committee plays a critical role in monitoring corporate accountability and the quality of financial reports (Carcello and Neal, 2000). For example, a larger audit committee is more likely to improve internal control quality, because the increased resources and enhanced status from more members will make the audit committee more effective in fulfilling its monitoring role (Zhang et al., 2007). In addition, a more independent audit committee will result in a higher quality of internal control (Krishnan, 2005). Hence, we further control for the size (*AUDIT_SIZE*) and independence (*AUDIT_INDEP*) of the audit committee in Models 4 and 5, respectively. Notably, the coefficient of *QFII_DUMMY* in both models remains positive and statistically significant, suggesting that our main result is not affected by the inclusion of audit committee variables.

4.3.3 Influence of domestic institutional investors

Domestic institutional investors affect both internal corporate governance and external monitoring (Ajinkya et al., 2005; Baik et al., 2010; Wang and Chen, 2017), which may affect the influence of QFIIs as foreign investors and institutional investors. Therefore, we additionally account for the influence of domestic institutional ownership (*INS_OWN*) to

determine whether QFIIs have a larger effect than domestic institutional investors. Otherwise, the positive coefficient of *QFII_DUMMY* may simply capture the effect of institutional investors. In Table 6, Model 6 reveals that *QFII_DUMMY* still attracts a significantly positive coefficient; however, the coefficient on *INS_OWN* is insignificant, reaffirming that QFIIs indeed matter to internal control quality and ruling out the alternative explanation.

4.3.4 Excluding companies in the manufacturing industry

In Table 1, Panel B shows that our observations are mainly clustered in the manufacturing industry, implying that our results may simply reflect industry effects. To further rule out such concerns, we ran Equation (1) by excluding the manufacturing sector. The results displayed in Model 7 of Table 6 confirm that our main finding is not driven by the manufacturing industry.

4.3.5 Dynamic system GMM approach

There are some common endogeneity concerns. For example, QFIIs may prefer to invest in companies with better internal control quality. The presence of QFIIs and internal control quality may be driven simultaneously by omitted variables. To address such concerns, we use the Arellano–Bond system GMM method, as suggested by Arellano and Bover (1995) and Blundell and Bond (1998), and include the one-year lagged internal control score as an independent variable in Equation (1).

The result from the dynamic panel system GMM approach presented in Model 1 of Table 7 shows that *QFII_DUMMY* has a significantly positive coefficient (t=3.5111). In addition, the Sargan test of over-identification has a p-value of less than 1%, and the Difference-in-Hansen test of exogeneity has a p-value of 0.151, which validates the

implementation of the dynamic panel data estimation. Simply put, the test presented in this section shows that our results are unlikely to be driven by potential endogeneity.¹³

[Insert Table 7 here]

4.3.6 PSM approach

The differences in observable fundamental characteristics between companies with and without QFIIs may result in biased estimations. Thus, we carry out a PSM approach to address this concern. With the calliper set at 0.001, we conducted a matching process with replacement and used the nearest neighbour technique to match each firm-year observation with QFIIs with a firm-year observation without QFIIs based on a battery of factors used as control variables in Equation (1). We then re-run Equation (1) based on the PSM sample to test the validity of H1.

Table 8 reports the results of the PSM analysis. Panel A shows that companies with QFIIs are associated with higher future internal control quality than companies without (ATT = 1.1196, t=2.39).¹⁴ The univariate results displayed in Panel B indicate that the sample is well balanced because the firm characteristics of the control group are not statistically different from those of the treatment group. The regression result based on the PSM sample is displayed in Panel C. The significantly positive coefficient on *QFII_DUMMY* suggests that our key finding is unlikely to be biased by the sample selection issue.

[Insert Table 8 here]

¹³ The GMM estimation could weaken the exogeneity assumption for an array of regressors, hence providing consistent estimates even if the reverse causality issue is present (Leszczensky and Wolbring, 2019). To further rule out the potential reverse causality, we follow Dyck et al. (2019) and Li et al. (2021) to implement the Granger causality test; specifically, we regress $QFII_DUMMY$ on lagged *ICI* and lagged $QFII_DUMMY$, with the same set of control variables included. The result is displayed in Appendix C. In Model 1 where the dependent variable is $QFII_DUMMY$, the estimate on LAG_ICI is insignificant, indicating that companies with high internal control quality are not likely to attract QFIIs. ¹⁴ Our results still hold if we conduct a matching process without replacement and are available upon request.

5. The mechanism influencing internal control

Key shareholders can influence corporate policies and business strategies through their voting power or shareholder activism (Lee and Lounsbury, 2011; Oh et al., 2011; Li et al., 2021). Similarly, if QFIIs are recognised as key shareholders, they may be entitled to significantly influence corporate policies (Huang and Zhu, 2015; Li et al., 2021). Thus, we further explored the mechanism through which QFIIs improve internal control quality. We introduce two categorical variables, *QFII_TOPTEN* and *QFII_NON_TOPTEN*, to capture the extent of the influence of QFIIs on internal control quality. *QFII_TOPTEN* equals one if the percentage of outstanding shares held by QFIIs is among the top ten largest shareholders of a company and zero otherwise. *QFII_NON_TOPTEN* equals one if the percentage of outstanding shares held by QFIIs is not among the top ten largest shareholders of a company and zero otherwise.

We then re-run Equation (1) by replacing *QFII_DUMMY* with *QFII_TOPTEN* and *QFII_NON_TOPTEN*. The results in Models 1 and 2 of Table 9 with *ICI* and *ADJ_ICI* as the dependent variables, respectively, show a positive and highly significant estimate of *QFII_TOPTEN* and an insignificant estimate of *QFII_NON_TOPTEN*. This means that the positive link between QFIIs and internal control quality becomes more salient if QFII is among the top ten shareholders. In Model 3 of Table 9, with *NO_DEFI* as the dependent variable, we find that the coefficients of *QFII_TOPTEN* and *QFII_TOPTEN* and *QFII_TOPTEN* are both significantly negative, and, more importantly, the magnitude of the estimate on *QFII_TOPTEN* is greater than that of the estimate on *QFII_NON_TOPTEN*. In sum, our evidence indicates that when QFIIs are the top ten largest shareholders, they may have a great scope to facilitate the effectiveness of internal control through their power and voice within investee companies.

[Insert Table 9 here]

6. Interplay between QFIIs and internal control quality on firm performance

Prior literature finds that higher quality of corporate internal control leads to superior financial performance (Stoel and Muhanna, 2011; Kuhn et al., 2013) and mitigates the loss of economic value in negative events (Wang et al., 2018). Hence, we posit that the improvement in internal control quality linked to QFIIs may lead to better financial performance. To test this conjecture, we employ the following model specification:

$$ROA_{i,t} = \alpha + \beta_1 QFII_DUMMY_{i,t-1} + \beta_2 QFII_DUMMY_{i,t-1} \times ICI_{i,t-1} + \beta_3 ICI_{i,t-1} + \beta_4 Control_{i,t-1} + Year FE + Industry FE + \varepsilon_{i,t}$$
(2)

where we use return on assets (*ROA*) to measure financial performance. The interaction term between *QFII_DUMMY* and *ICI* captures the incremental effect of internal control quality through the presence of QFII-licensed investors on firm performance. We expect the estimate of the interaction term to be significantly positive. In addition, we control for *FIRMSIZE*, *LEVERAGE*, *OCF*, *SOE*, *GROWTH*, and *BIG4*. In Table 10, Model 1 shows significantly positive estimates of *QFII_DUMMY* and *ICI*, indicating that QFIIs and better internal control quality will improve future operating performance. More importantly, in Model 2, the interaction term, *QFII_DUMMY* × *ICI*, has a significantly positive coefficient, suggesting that the increase in internal control quality attributable to QFIIs results in better performance.

[Insert Table 10 here]

7. Conclusions

In this study, we investigate the role of QFIIs in companies' internal control quality in the Chinese context. Using a large sample between 2005 and 2017, we provide robust evidence that companies with QFIIs experience an increase in future internal control scores and exhibit fewer internal control deficiencies. Moreover, QFII ownership is positively linked to the quality of internal control. This positive influence is more pronounced in companies with QFIIs than countries with high institutional quality, which is consistent with their motives. Our findings remained robust throughout robustness tests, including using alternative measures of key variables, controlling for additional effects of the size of the audit committee, the audit committee independence, and domestic institutional ownership, as well as a subsample excluding manufacturing companies. Our findings continue to hold after applying dynamic GMM and PSM methods. Further, the positive influence of QFIIs on internal control quality appears more salient when the QFII is among the top ten largest shareholders, which provides foreign investors with sufficient voting power to push companies to induce the construction of an effective internal control system. Finally, we find that the improvement in internal control quality attributable to QFIIs drives firm performance.

Our study also opens up a potential avenue for future research. The CSRC extensively enacted a series of guidelines and reforms to relax the QFII selection criteria to attract more foreign capital to the domestic Chinese equity markets after 2016. Later in 2019, the CSRC announced simplified rules that removed the relevant criteria of assets under management and years of experience for the foreign investors. As such, one may argue that this relaxation could raise concerns on the quality of the newly joined qualified foreign investors, potentially deteriorating the subsequent internal control quality. However, given that the majority of the newly approved QFIIs (i.e., the Vanguard Group, J.P. Morgan Securities, Nomura Limited, and Marshall Wace) by the Chinese regulatory authorities are reputable and top-tier financial institutions and still subject to a high level of governance standards in their home countries and securities commissions,¹⁵ we conjecture that they still have motives and ability to transplant their high-quality governance practices and awareness to their investee companies in China, consistently pushing towards internal control quality. Moreover, given the mounting empirical evidence that QFIIs have consistently helped enhance the corporate governance of Chinese listed firms for decades (Huang and Zhu, 2015; Cao et al., 2017; Kim et al., 2020), the reforms

¹⁵ For details, please see <u>http://www.csrc.gov.cn/pub/csrc_en/OpeningUp/RelatedLists/QFIIs/</u>.

after 2016 are less likely to distort the positive influence of QFIIs on their investee companies. Due to the data availability, we are unable to empirically examine the role of QFIIs in corporate governance practices after these recent reforms. Hence, it would be interesting for future studies to see whether the relaxation of the restrictions on the QFII qualification and selection criteria may play a part in the interplay between QFIIs and internal control.

Overall, this study offers valuable implications for company-level stakeholders and regulatory authorities. Our findings indicate that foreign institutional investors, especially those from countries with high institutional quality, could exert a positive influence on the internal control systems of investee companies. In addition, QFIIs may facilitate the attainment of broader economic goals by constructing an efficient internal control system. Thus, other developing economies sharing similar institutional backgrounds could consider introducing a similar scheme that attracts foreign investment institutions, to help their domestic companies achieve higher internal governance quality and economic outcomes.

References

- Aggarwal, R., Erel, I., Ferreira, M. & Matos, P. (2011). Does governance travel around the world? Evidence from institutional investors. *Journal of Financial Economics*, 100(1),154-181.
- Ajinkya, B., Bhojraj, S. & Sengupta, P. (2005). The Association between Outside Directors, Institutional Investors and the Properties of Management Earnings Forecasts. *Journal* of Accounting Research, 43(3),343–376.
- Arellano, M. & Bover, O. (1995). Another look at the instrumental variable estimation of errorcomponents models. *Journal of Econometrics*, 68(1),29-51.
- Ashbaugh-Skaife, H., Collins, D. W., Kinney JR, W. R. & Lafond, R. (2009). The Effect of SOX Internal Control Deficiencies on Firm Risk and Cost of Equity. *Journal of Accounting Research*, 47(1),1-43.
- Ashbaugh-Skaife, H., Collins, D. W. & Kinney, W. R. (2007). The discovery and reporting of internal control deficiencies prior to SOX-mandated audits. *Journal of Accounting and Economics*, 44(1),166-192.
- Bai, C.-E., Liu, Q., Lu, J., Song, F. M. & Zhang, J. (2004). Corporate governance and market valuation in China. *Journal of Comparative Economics*, 32(4),599-617.
- Baik, B., Kang, J.-K. & Kim, J.-M. (2010). Local institutional investors, information asymmetries, and equity returns. *Journal of Financial Economics*, 97(1),81-106.
- Bardhan, I., Lin, S. & Wu, S.-L. (2015). The Quality of Internal Control over Financial Reporting in Family Firms. *Accounting Horizons*, 29(1),41-60.
- Blundell, R. & Bond, S. (1998). Initial conditions and moment restrictions in dynamic panel data models. *Journal of Econometrics*, 87(1),115-143.
- Cao, L., Du, Y. & Hansen, J. Ø. (2017). Foreign institutional investors and dividend policy: Evidence from China. *International Business Review*, 26(5),816-827.
- Carcello, J. V. & Neal, T. L. (2000). Audit Committee Composition and Auditor Reporting. *The Accounting Review*, 75(4),453-467.
- Chalmers, K., Hay, D. & Khlif, H. (2019). Internal control in accounting research: A review. *Journal of Accounting Literature*, 42,80-103.
- Changchit, C., Holsapple, C. W. & Viator, R. E. (2001). Transferring auditors' internal control evaluation knowledge to management. *Expert Systems With Applications*, 20(3),275-291.
- Chen, G. Z. & Keung, E. C. (2018). Corporate diversification, institutional investors and internal control quality. *Accounting & Finance*, 53(3),751-786.
- Chen, J., Chan, K. C., Dong, W. & Zhang, F. F. (2017). Internal Control and Stock Price Crash Risk: Evidence from China. *European Accounting Review*, 26(1),125-152.
- Cheng, M., Dhaliwal, D. & Zhang, Y. (2013). Does investment efficiency improve after the disclosure of material weaknesses in internal control over financial reporting? *Journal of Accounting and Economics*, 56(1),1-18.
- Del Bosco, B. & Misani, N. (2016). The effect of cross-listing on the environmental, social, and governance performance of firms. *Journal of World Business*, 51(6),977-990.
- Deumes, R. W. J. & Knechel, W. R. (2008). Economic Incentives for Voluntary Reporting on Internal Risk Management and Control Systems. *Auditing: A Journal of Practice and Theory*, 27(1),35-66.
- DiMaggio, P. J. & Powell, W. W. (1983). The Iron Cage Revisited: Institutional Isomorphism and Collective Rationality in Organizational Fields. *American Sociological Review*, 48(2),147-160.

- Douma, S., George, R. & Kabir, R. (2006). Foreign and domestic ownership, business groups, and firm performance: Evidence from a large emerging market. *Strategic Management Journal*, 27(7),637-657.
- Doyle, J. T., Ge, W. & McVay, S. (2007). Accruals Quality and Internal Control over Financial Reporting. *The Accounting Review*, 82(5),1141-1170.
- Dyck, A., Lins, K. V., Roth, L. & Wagner, H. F. (2019). Do institutional investors drive corporate social responsibility? International evidence. *Journal of Financial Economics*, 131(3),693-714.
- Fang, V. W., Maffett, M. & Zhang, B. (2015). Foreign Institutional Ownership and the Global Convergence of Financial Reporting Practices. *Journal of Accounting Research*, 53(3),593-631.
- Feng, M., Li, C. & McVay, S. (2009). Internal control and management guidance. *Journal of Accounting and Economics*, 48(2),190-209.
- Ferreira, M. A. & Matos, P. (2008). The colors of investors' money: The role of institutional investors around the world. *Journal of Financial Economics*, 88(3),499-533.
- Gong, G., Ke, B. & Yu, Y. (2013). Home Country Investor Protection, Ownership Structure and Cross-Listed Firms' Compliance with SOX-Mandated Internal Control Deficiency Disclosures. *Contemporary Accounting Research*, 30(4),1490-1523.
- Gul, F. A., Kim, J.-B. & Qiu, A. A. (2010). Ownership concentration, foreign shareholding, audit quality, and stock price synchronicity: Evidence from China. *Journal of Financial Economics*, 95(3),425-442.
- Han, L., Zheng, Q., Li, L. & Yin, L. (2015). Do foreign institutional investors stabilize the capital market? *Economics Letters*, 136,73-75.
- He, X., Rui, O., Zheng, L. & Zhu, H. (2014). Foreign ownership and auditor choice. *Journal* of Accounting and Public Policy, 33(4),401-418.
- Hu, G., Yuan, R. & Xiao, J. Z. (2017). Can independent directors improve internal control quality in China? *The European Journal of Finance*, 23(7-9),626-647.
- Huang, R. D. & Shiu, C. Y. (2009). Local effects of foreign ownership in an emerging financial market: Evidence from qualified foreign institutional investors in Taiwan. *Financial Management*, 38(3),567-602.
- Huang, W. & Zhu, T. (2015). Foreign institutional investors and corporate governance in emerging markets: Evidence of a split-share structure reform in China. *Journal of Corporate Finance*, 32,312-326.
- Ji, X.-d., Lu, W. & Qu, W. (2017). Voluntary Disclosure of Internal Control Weakness and Earnings Quality: Evidence From China. *The International Journal of Accounting*, 52(1),27-44.
- Jia, N., Shi, J., Wang, C. & Wang, Y. (2020). Parasites and Paragons: Ownership Reform and Concentrated Interest among Minority Shareholders. *Journal of Management Studies*, 57(1),129-162.
- Jiang, F. & Kim, K. A. (2015). Corporate governance in China: A modern perspective. *Journal* of Corporate Finance, 32,190-216.
- Jin, D., Wang, H., Wang, P. & Yin, D. (2016). Social trust and foreign ownership: Evidence from qualified foreign institutional investors in China. *Journal of Financial Stability*, 23,1-14.
- Kang, J.-K. & Kim, J.-M. (2010). Do foreign investors exhibit a corporate governance disadvantage? An information asymmetry perspective. *Journal of International Business Studies*, 41(8),1415-1438.
- Kim, J.-B., Li, X., Luo, Y. & Wang, K. (2020). Foreign Investors, External Monitoring, and Stock Price Crash Risk. *Journal of Accounting, Auditing & Finance*, 35(4),829-853.

- Kim, J.-B., Pevzner, M. & Xin, X. (2019). Foreign institutional ownership and auditor choice: Evidence from worldwide institutional ownership. *Journal of International Business Studies*, 50(1),83-111.
- Krishnan, J. (2005). Audit Committee Quality and Internal Control: An Empirical Analysis. *The Accounting Review*, 80(2),649-675.
- Kuhn, J. R., Ahuja, M. & Mueller, J. (2013). An examination of the relationship of IT control weakness to company financial performance and health. *International Journal of Accounting & Information Management*, 21(3),227-240.
- La Porta, R., Lopez-de-Silanes, F. & Shleifer, A. (2008). The economic consequences of legal origins. *Journal of Economic Literature*, 46(2),285-332.
- Lee, M.-D. P. & Lounsbury, M. (2011). Domesticating radical rant and rage: An exploration of the consequences of environmental shareholder resolutions on corporate environmental performance. *Business & Society*, 50(1),155-188.
- Lel, U. (2019). The role of foreign institutional investors in restraining earnings management activities across countries. *Journal of International Business Studies*, 50(6),895-922.
- Leszczensky, L. & Wolbring, T. (2019). How to Deal With Reverse Causality Using Panel Data? Recommendations for Researchers Based on a Simulation Study. *Sociological Methods & Research*.
- Li, W., Han, Y. & He, J. (2019). How Does the Heterogeneity of Internal Control Weakness Affect R&D Investment? *Emerging Markets Finance and Trade*, 55(15),3591-3614.
- Li, Z., Wang, P. & Wu, T. (2021). Do foreign institutional investors drive corporate social responsibility? Evidence from listed firms in China. *Journal of Business Finance & Accounting*, 48(1-2),338-373.
- Liu, J., Zhang, D., Cai, J. & Davenport, J. (2019). Legal Systems, National Governance and Renewable Energy Investment: Evidence from Around the World. *British Journal of Management*.
- Liu, N., Bredin, D., Wang, L. & Yi, Z. (2014). Domestic and foreign institutional investors' behavior in China. *The European Journal of Finance*, 20(7-9),728-751.
- Lu, Y. & Cao, Y. (2018). The individual characteristics of board members and internal control weakness: Evidence from China. *Pacific-Basin Finance Journal*, 51,75-94.
- Luong, H., Moshirian, F., Nguyen, L., Tian, X. & Zhang, B. (2017). How do foreign institutional investors enhance firm innovation? *Journal of Financial and Quantitative Analysis*, 52(4),1449-1490.
- Maijoor, S. (2000). The Internal Control Explosion. *International Journal of Auditing*, 4(1),101-109.
- Oh, W. Y., Chang, Y. K. & Martynov, A. (2011). The effect of ownership structure on corporate social responsibility: Empirical evidence from Korea. *Journal of Business Ethics*, 104(2),283-297.
- Oradi, J., Asiaei, K. & Rezaee, Z. (2020). CEO financial background and internal control weaknesses. *Corporate Governance: An International Review*, 28(2),119-140.
- Petersen, M. A. (2009). Estimating standard errors in finance panel data sets: Comparing approaches. *The Review of Financial Studies*, 22(1),435-480.
- Schuppli, M. & Bohl, M. T. (2010). Do foreign institutional investors destabilize China's Ashare markets? *Journal of International Financial Markets, Institutions and Money*, 20(1),36-50.
- Stoel, M. D. & Muhanna, W. A. (2011). IT internal control weaknesses and firm performance: An organizational liability lens. *International Journal of Accounting Information* Systems, 12(4),280-304.
- Thompson, S. B. (2011). Simple formulas for standard errors that cluster by both firm and time. *Journal of Financial Economics*, 99(1),1-10.

- Wang, F., Xu, L., Zhang, J. & Shu, W. (2018). Political connections, internal control and firm value: Evidence from China's anti-corruption campaign. *Journal of Business Research*, 86,53-67.
- Wang, M. & Chen, Y. (2017). Does voluntary corporate social performance attract institutional investment? Evidence from China. *Corporate Governance: An International Review*, 25(5),338-357.
- Wu, L. & Yue, H. (2009). Corporate tax, capital structure, and the accessibility of bank loans: Evidence from China. *Journal of Banking and Finance*, 33(1),30-38.
- Yu, J., Campbell, S. M. R., Li, J. & Zhang, Z. (2019). Do sources of occupational community impact corporate internal control? The case of CFOs in the high-tech industry. *Accounting, Auditing & Accountability Journal*, 32(4),957-983.
- Yuan, R., Xiao, J. Z., Milonas, N. & Zou, J. H. (2009). The Role of Financial Institutions in the Corporate Governance of Listed Chinese Companies. *British Journal of Management*, 20(4),562-581.
- Zhang, X., Yang, X., Strange, R. & Zhang, Q. (2017). Informed trading by foreign institutional investors as a constraint on tunneling: Evidence from China. *Corporate Governance: An International Review*, 25(4),222-235.
- Zhang, Y., Zhou, J. & Zhou, N. (2007). Audit committee quality, auditor independence, and internal control weaknesses. *Journal of Accounting and Public Policy*, 26(3),300-327.

Table 1 Sample distributionPanel A: Annual distribution

Year	QFII_DUMMY=1	QFII_DUMMY=0	No. of Obs.
2005	13	517	530
2006	74	849	923
2007	158	996	1,154
2008	127	1,108	1,235
2009	106	1,258	1,364
2010	179	1,300	1,479
2011	178	1,392	1,570
2012	123	1,812	1,935
2013	142	2,086	2,228
2014	195	2,180	2,375
2015	224	2,129	2,353
2016	169	2,312	2,481
2017	268	2,415	2,683
Total	1,956	20,354	22,310

Panel B: Distribution by industry

CSRC industry classification	QFII_DUMMY	QFII_DUMMY	No. of Obs.
	= 1	= 0	
A: Agriculture, Forestry, Animal Husbandry and Fishery	26	317	343
B: Mining	69	604	673
C: Manufacturing	1,198	11,807	13,005
D: Production and Supply of Electric Power, Heat Power, Gas and Water	68	943	1,011
E: Construction	51	529	580
F: Wholesale and Retail Trade	112	1,407	1,519
G: Transport, Storage and Postal Services	159	688	847
H: Accommodation and Catering Service	12	95	107
I: Information Transmission, Software and Information Technology Services	71	1,295	1,366
K: Real Estate	84	1,252	1,336
L: Leasing and Business Services	20	328	348
M: Scientific Research and Technical Services	9	152	161
N: Water Conservancy, Environment and. Public Facilities Management	37	283	320
P: Education	0	48	48
Q: Health and Social Work	6	83	89
R: Culture, Sports and Entertainment	27	279	306
S: Comprehensive (Miscellaneous)	7	244	251
Total	1,956	20,354	22,310

Variable	Ν	Mean	SD	Min	P25	P50	P75	Max
ICI	22,310	64.7354	14.9141	0.0000	61.9920	67.6815	71.1550	99.5360
ADJ_ICI	22,310	-2.4127	14.5454	-78.3650	-4.3060	0.0000	3.6600	68.1430
NO_DEFI	20,145	0.8299	23.7366	0.0000	0.0000	0.0000	0.0000	2015.0000
RESTATEMENT	24,285	0.2039	0.4029	0.0000	0.0000	0.0000	0.0000	1.0000
NO_RESTATEMENT	24,285	0.2595	0.5946	0.0000	0.0000	0.0000	0.0000	11.0000
QFII_DUMMY	22,310	0.0877	0.2828	0.0000	0.0000	0.0000	0.0000	1.0000
QFII_OWN (QFII_DUMMY=1)	1,956	1.9415	2.2767	0.0115	0.5691	1.1914	2.3457	20.9283
NO_QFII	22,310	0.0731	0.2494	0.0000	0.0000	0.0000	0.0000	2.0794
QFII_HIGHWGI	22,310	0.0435	0.2039	0.0000	0.0000	0.0000	0.0000	1.0000
QFII_LOWWGI	22,310	0.0442	0.2055	0.0000	0.0000	0.0000	0.0000	1.0000
QFII_TOPTEN	22,310	0.0702	0.2555	0.0000	0.0000	0.0000	0.0000	1.0000
QFII_NON_TOPTEN	22,310	0.0175	0.1311	0.0000	0.0000	0.0000	0.0000	1.0000
FIRMSIZE	22,310	21.8420	1.3301	10.8422	20.9555	21.7060	22.5754	28.5087
FIRMAGE	22,310	2.1506	0.6883	0.0000	1.6094	2.3026	2.7081	3.2958
LEVERAGE	22,310	0.4734	0.2294	0.0532	0.3013	0.4716	0.6305	1.2988
GROWTH	22,310	0.2185	0.5893	-0.7188	-0.0283	0.1165	0.2941	3.9429
LOSS	22,310	0.1092	0.3119	0.0000	0.0000	0.0000	0.0000	1.0000
OWN_CON	22,310	0.1812	0.1783	0.0011	0.0309	0.1101	0.3006	0.6613
SOE	22,310	0.4864	0.4998	0.0000	0.0000	0.0000	1.0000	1.0000
BOARDSIZE	22,310	2.2789	0.1824	1.3863	2.1972	2.3026	2.3026	2.9957
INDEP	22,310	0.3675	0.0527	0.0000	0.3333	0.3333	0.4000	0.5714
DUALITY	22,310	0.2113	0.4083	0.0000	0.0000	0.0000	0.0000	1.0000
MEETING	22,310	2.2864	0.3435	0.6931	2.0794	2.3026	2.4849	4.0604
BIG4	22,310	0.0607	0.2388	0.0000	0.0000	0.0000	0.0000	1.0000
AUDIT_SIZE	12,120	0.2862	0.5853	0.0000	0.0000	0.0000	0.0000	2.0794
AUDIT_INDEP	12,120	0.1269	0.2681	0.0000	0.0000	0.0000	0.0000	1.0000
INS_OWN (%)	22,272	48.1909	22.5142	0.0661	32.7600	50.6264	65.2226	92.4232
ROA	22,281	0.0480	0.0738	-0.3613	0.0250	0.0476	0.0782	0.2475
OCF	22,281	0.0505	0.0918	-0.2648	0.0036	0.0486	0.0990	0.3458

Note: All variables are defined in Appendix A.

	Variable	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
(1)	ICI	1.0000								
(2)	ADJ_ICI	0.9758*	1.0000							
(3)	NO_DEFI	0.0109	0.0133*	1.0000						
(4)	QFII_DUMMY	0.0966*	0.0943*	-0.0040	1.0000					
(5)	QFII_HIGHWGI	0.0830*	0.0819*	-0.0045	0.6877*	1.0000				
(6)	QFII_LOWWGI	0.0506*	0.0485*	-0.0011	0.6937*	-0.0458*	1.0000			
(7)	FIRMSIZE	0.3192*	0.3133*	0.0439*	0.1530*	0.1341*	0.0775*	1.0000		
(8)	FIRMAGE	-0.1212*	-0.1113*	0.0060	0.0300*	0.0247*	0.0168*	0.2193*	1.0000	
(9)	LEVERAGE	-0.2038*	-0.2082*	0.0140*	-0.0165*	-0.0064	-0.0164*	0.2406*	0.3492*	1.0000
(10)	GROWTH	-0.0718*	-0.0696*	-0.0040	-0.0038	-0.0055	0.0003	-0.0868*	-0.0146*	0.0094
(11)	LOSS	-0.4504*	-0.4339*	-0.0046	-0.0486*	-0.0366*	-0.0306*	-0.1264*	0.1161*	0.2469*
(12)	OWN_CON	0.0356*	0.0439*	0.0161*	0.0646*	0.0591*	0.0303*	0.3542*	0.2974*	0.0835*
(13)	SOE	0.0737*	0.0577*	0.0269*	0.0747*	0.0559*	0.0473*	0.2754*	0.3376*	0.2272*
(14)	BOARDSIZE	0.1185*	0.1021*	0.0283*	0.0560*	0.0426*	0.0347*	0.2387*	0.0579*	0.1172*
(15)	INDEP	-0.0120*	-0.0044	-0.0026	-0.0014	-0.0007	-0.0012	0.0380*	-0.0252*	-0.0235*
(16)	DUALITY	-0.0351*	-0.0286*	-0.0073	-0.0168*	-0.0145*	-0.0088	-0.1288*	-0.1792*	-0.1175*
(17)	MEETING	0.0022	0.0116*	0.0336*	-0.0183*	-0.0093	-0.0160*	0.2043*	0.0553*	0.1271*
(18)	BIG4	0.1390*	0.1301*	0.0702*	0.1177*	0.0995*	0.0632*	0.3408*	0.0584*	0.0563*
		(10)	(11)	(12)	(13)	(14)	(15)	(16)	(17)	(18)
(10)	GROWTH	1.0000								
(11)	LOSS	0.0380*	1.0000							
(12)	OWN_CON	-0.0507*	-0.0145*	1.0000						
(13)	SOE	-0.0606*	0.0387*	0.1975*	1.0000					
(14)	BOARDSIZE	-0.0495*	-0.0266*	0.0256*	0.2762*	1.0000				
(15)	INDEP	0.0204*	0.0041	0.0581*	-0.1010*	-0.4383*	1.0000			
(16)	DUALITY	0.0210*	-0.0141*	-0.0903*	-0.2656*	-0.1712*	0.1033*	1.0000		
(17)	MEETING	0.0575*	-0.0188*	0.0133*	-0.1007*	-0.0569*	0.0685*	0.0249*	1.0000	
(18)	BIG4	-0.0229*	-0.0312*	0.1369*	0.1294*	0.1188*	0.0215*	-0.0649*	0.0247*	1.0000

 Table 3 Correlation matrix

Note: The * indicates statistical significance at least the 0.1 level. All variables are defined in Appendix A.

Table 4	Effect of	OFIIs on	internal	control	quality
14010 1	Direct or	XI IID OII	meentai	001101	quanty

Dep. Var. =	ICI	ICI	ADJ_ICI	ADJ_ICI	NO_DEFI	NO_DEFI
	(1)	(2)	(3)	(4)	(5)	(6)
	Baseline		Industry-median-	adjusted internal	The number of	internal control
			control	index	deficiencies (weaknesses)
QFII_DUMMY	0.5105**		0.6234**		-1.0521**	
	(1.9780)		(2.4123)		(-2.2721)	
QFII_OWN		0.2569***		0.2996***		-0.2225**
		(3.4461)		(4.0268)		(-1.9773)
FIRMSIZE	4.6356***	4.6369***	4.3426***	4.3451***	-0.0088	-0.0307
	(43.6172)	(43.8109)	(40.6417)	(40.8535)	(-0.0741)	(-0.2567)
FIRMAGE	-1.5967***	-1.5962***	-1.4447***	-1.4439***	-0.1765	-0.1785
	(-10.7471)	(-10.7457)	(-9.7805)	(-9.7774)	(-1.1008)	(-1.1091)
LEVERAGE	-15.6709***	-15.6605***	-14.9712***	-14.9613***	0.4963	0.5433
	(-26.5289)	(-26.5081)	(-25.1047)	(-25.0895)	(0.9721)	(1.0559)
GROWTH	-0.4737**	-0.4771**	-0.4906**	-0.4944**	-0.1092	-0.1097
	(-2.0871)	(-2.1018)	(-2.1648)	(-2.1813)	(-0.8553)	(-0.8580)
LOSS	-16.0203***	-16.0086***	-15.3133***	-15.3001***	-0.3541	-0.3551
	(-38.2103)	(-38.1875)	(-36.3339)	(-36.3081)	(-1.3471)	(-1.3499)
OWN_CON	2.8618***	2.9145***	3.1204***	3.1832***	-0.5449	-0.6221
	(5.1117)	(5.2024)	(5.5474)	(5.6561)	(-0.7097)	(-0.7915)
SOE	-0.2104	-0.2042	-0.3095	-0.3021	0.9317**	0.9197**
	(-1.0526)	(-1.0214)	(-1.5442)	(-1.5070)	(2.4523)	(2.4473)
BOARDSIZE	0.6041	0.5771	0.3925	0.3612	2.2915***	2.2944***
	(1.1328)	(1.0820)	(0.7309)	(0.6727)	(2.8828)	(2.8832)
INDEP	1.8002	1.7571	1.3728	1.3232	0.8556	0.8535
	(1.0486)	(1.0235)	(0.7949)	(0.7661)	(0.5208)	(0.5198)
DUALITY	0.0563	0.0503	0.0689	0.0622	0.1864	0.1842
	(0.2683)	(0.2396)	(0.3288)	(0.2970)	(0.5726)	(0.5657)
MEETING	-0.2797	-0.2841	-0.0649	-0.0706	2.4242	2.4400
	(-1.0720)	(-1.0889)	(-0.2476)	(-0.2693)	(1.6212)	(1.6264)
BIG4	0.1585	0.1575	0.1515	0.1523	6.4168**	6.3716**
	(0.4438)	(0.4426)	(0.4291)	(0.4328)	(2.4330)	(2.4322)
_constant	-21.5041***	-21.4669***	-80.2283***	-80.2030***	-11.8941***	-11.5189***
	(-8.8361)	(-8.8424)	(-32.6555)	(-32.7301)	(-2.9275)	(-2.9386)
Year-FE	Yes	Yes	Yes	Yes	Yes	Yes
Industry-FE	Yes	Yes	Yes	Yes	Yes	Yes
No. of obs.	22,310	22,310	22,310	22,310	20,145	20,145
Adj. R-square	0.380	0.380	0.345	0.346	0.010	0.009

Notes: This table reports the results of OLS regressions of the influence of QFIIs on companies' internal control index and the number of reported internal control deficiencies. The dependent variable is the overall internal control index score (*ICI*) in Models (1) and (2), the industry-median-adjusted internal control score (*ADJ_ICI*) in Models (3) and (4), and the number of reported internal control deficiencies (*NO_DEFI*) in Models (5) and (6). All variables are defined in Appendix A. In all specifications, independent variables except *GROWTH* and *LOSS* are lagged by one year. Industry and year dummies are included in each regression. We cluster standard errors by year and firm. The 0.01, 0.05, and 0.1 significance levels are denoted by ***, **, and * (two-tailed), respectively.

Dep. Var. =	ICI	ICI	ADJ_ICI	ADJ_ICI	NO_DEFI	NO_DEFI
	(1)	(2)	(3)	(4)	(5)	(6)
QFII_HIGHWGI	0.9966***		1.1560***		-1.4726***	
	(2.9256)		(3.3850)		(-2.7170)	
QFII_LOWWGI	0.0405		0.1084		-0.6626	
	(0.1131)		(0.3023)		(-1.5287)	
QFII_HIGHWGI_OWN		0.3447***		0.3920***		-0.2688*
		(3.7034)		(4.0856)		(-1.8884)
QFII_LOWWGI_OWN		0.1366		0.1732		-0.1662
-		(1.0657)		(1.3803)		(-1.4325)
FIRMSIZE	4.6316***	4.6358***	4.3382***	4.3439***	-0.0044	-0.0299
	(43.5489)	(43.7918)	(40.5655)	(40.8321)	(-0.0369)	(-0.2507)
FIRMAGE	-1.5962***	-1.5966***	-1.4441***	-1.4442***	-0.1760	-0.1782
	(-10.7430)	(-10.7475)	(-9.7764)	(-9.7795)	(-1.0984)	(-1.1074)
LEVERAGE	-15.6686***	-15.6597***	-14.9687***	-14.9604***	0.4944	0.5428
	(-26.5224)	(-26.5049)	(-25.0974)	(-25.0861)	(0.9686)	(1.0550)
GROWTH	-0.4732**	-0.4761**	-0.4901**	-0.4933**	-0.1089	-0.1098
	(-2.0851)	(-2.0972)	(-2.1627)	(-2.1765)	(-0.8530)	(-0.8583)
LOSS	-16.0208***	-16.0108***	-15.3139***	-15.3024***	-0.3540	-0.3543
	(-38.2132)	(-38.1888)	(-36.3371)	(-36.3099)	(-1.3467)	(-1.3478)
OWN_CON	2.8520***	2.9136***	3.1097***	3.1823***	-0.5385	-0.6227
	(5.0933)	(5.2008)	(5.5274)	(5.6544)	(-0.7024)	(-0.7921)
SOE	-0.2081	-0.2025	-0.3070	-0.3003	0.9295**	0.9189**
	(-1.0413)	(-1.0130)	(-1.5319)	(-1.4981)	(2.4493)	(2.4465)
BOARDSIZE	0.6091	0.5794	0.3980	0.3636	2.2850***	2.2939***
	(1.1424)	(1.0864)	(0.7414)	(0.6773)	(2.8785)	(2.8828)
INDEP	1.8097	1.7569	1.3833	1.3229	0.8326	0.8531
	(1.0544)	(1.0233)	(0.8012)	(0.7659)	(0.5064)	(0.5196)
DUALITY	0.0561	0.0493	0.0687	0.0611	0.1877	0.1847
	(0.2674)	(0.2346)	(0.3279)	(0.2917)	(0.5765)	(0.5673)
MEETING	-0.2792	-0.2836	-0.0644	-0.0701	2.4229	2.4395
	(-1.0703)	(-1.0869)	(-0.2456)	(-0.2673)	(1.6205)	(1.6261)
BIG4	0.1508	0.1594	0.1432	0.1544	6.4274**	6.3717**
	(0.4226)	(0.4482)	(0.4056)	(0.4388)	(2.4351)	(2.4321)
_constant	-21.4331***	-21.4521***	-80.1505***	-80.1874***	-11.9725***	-11.5347***
	(-8.8041)	(-8.8355)	(-32.6113)	(-32.7198)	(-2.9360)	(-2.9402)
Year-FE	Yes	Yes	Yes	Yes	Yes	Yes
Industry-FE	Yes	Yes	Yes	Yes	Yes	Yes
No. of obs.	22,310	22,310	22,310	22,310	20,145	20,145
Adj. R-square	0.380	0.380	0.345	0.346	0.009	0.009

Table 5 Effect of the institutional quality of QFIIs' home countries on internal control quality

Notes: This table displays the results of the influence of the institutional quality of QFIIs' countries on *ICI*, *ADJ_ICI*, and *NO_DEFI*. All variables are defined in Appendix A. In all specifications, independent variables except *GROWTH* and *LOSS* are lagged by one year. Industry and year dummies are included in each regression. We cluster standard errors by year and firm. The 0.01, 0.05, and 0.1 significance levels are denoted by ***, **, and * (two-tailed), respectively.

Table 6 Robustness tests

Dep. Var. =	RESTATEMENT	NO_RESTATEMENT	ICI	ICI	ICI	ICI	ICI
	The probability of a	The number of	Number of QFIIs	Characteristics of	f audit committee	Domestic institutional	Excluding
	restatement	restatements	-			investors	manufacturing firms
	(1)	(2)	(3)	(4)	(5)	(6)	(7)
QFII_DUMMY	-0.0585*	-0.0246**		0.9281**	0.9233**	0.5070**	0.7794*
	(-1.6453)	(-2.0710)		(2.3078)	(2.2969)	(1.9635)	(1.7996)
NO_QFII			1.2367***	· · · ·	· · · ·		
			(4.2243)				
AUDIT SIZE				0.3128			
—				(1.1046)			
AUDIT INDEP				. ,	0.4692		
—					(0.7724)		
INS OWN					· · · ·	0.0018	
_						(0.4140)	
FIRMSIZE	-0.0413***	-0.0181***	3.9678***	2.4945***	2.4960***	4.6231***	3.8746***
	(-4.4176)	(-4.6015)	(37.4310)	(16.5010)	(16.5179)	(42.7153)	(24.4585)
FIRMAGE	0.0111	0.0043	-2.8956***	-2.7232***	-2.7139***	-1.5983***	-2.4391***
	(0.8167)	(0.8028)	(-19.0338)	(-12.2074)	(-12.1727)	(-10.6592)	(-10.6215)
LEVERAGE	0.2316***	0.1032***	-0.0633	-1.7761*	-1.7763*	-15.6365***	-0.0382
	(4.7716)	(4.8031)	(-1.2813)	(-1.7761)	(-1.7757)	(-26.4390)	(-1.4862)
GROWTH	0.0389**	0.0165**	-0.6012**	-0.3733	-0.3740	-0.4842**	-1.0742***
	(2.4530)	(2.0223)	(-2.4942)	(-1.1305)	(-1.1324)	(-2.1304)	(-3.2796)
LOSS	0.2475***	0.1287***	-18.7591***	-16.8665***	-16.8650***	-16.0292***	-18.7661***
	(7.9491)	(7.5211)	(-43.9338)	(-24.2158)	(-24.2154)	(-38.1910)	(-26.9981)
OWN CON	-0.1725**	-0.0873***	2.3347***	3.4592***	3.4615***	2.7980***	1.2917
_	(-2.5644)	(-3.3720)	(4.1022)	(4.5879)	(4.5897)	(4.5767)	(1.4895)
SOE	-0.0810***	-0.0366***	-0.2502	-0.6685**	-0.6602**	-0.2255	0.3094
	(-3.5308)	(-3.9545)	(-1.2162)	(-2.0320)	(-2.0096)	(-1.1042)	(0.9659)
BOARDSIZE	0.0518	0.0181	0.1821	0.0691	0.0928	0.5913	-0.0684
	(0.8243)	(0.7237)	(0.3297)	(0.0763)	(0.1025)	(1.1071)	(-0.0807)
INDEP	0.0628	0.0693	1.0581	7.8241***	7.8214***	1.7802	-5.0659*
	(0.3086)	(0.8440)	(0.5939)	(3.0227)	(3.0216)	(1.0344)	(-1.7584)
DUALITY	0.0268	0.0053	0.0373	-0.2260	-0.2275	0.0520	-0.6378
	(1.1543)	(0.5396)	(0.1713)	(-0.8134)	(-0.8185)	(0.2466)	(-1.5845)
MEETING	0.1314***	0.0506***	-1.3433***	-1.9860***	-1.9873***	-0.2654	-1.1980***
	(4.4487)	(4.2917)	(-5.0168)	(-5.1713)	(-5.1754)	(-1.0168)	(-3.0089)
BIG4	-0.1592***	-0.0455***	0.8455**	0.9060*	0.9068*	0.1540	0.9992*
	(-3.4775)	(-3.2751)	(2.3108)	(1.6555)	(1.6552)	(0.4296)	(1.9276)
constant	0.0412	0.6782***	-8.5221***	18.4187***	18.3129***	-21.3425***	-4.8034
—	(0.1652)	(6.4581)	(-3.4757)	(4.9280)	(4.9002)	(-8.7495)	(-1.3247)
Year-FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Industry-FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes
No. of obs.	24,285	24,285	22,310	12,120	12,120	22,272	9,305
Adi./Pseudo R-square	0.0328	0.0322	0.340	0.231	0.231	0.379	0.351

Notes: In Models (1) and (2), the dependent variable is the probability of a financial restatement (*RESTATEMENT*) and the number of restatements (*NO_RESTATEMENT*), respectively. In Model (3), the key independent variable is the number of QFIIs (*NO_QFII*) of a company in a fiscal year, and *NO_QFII* is in the natural logarithm form. Models (4) and (5) controls for the characteristics of the audit committee (*AUDIT_SIZE* and *AUDIT_INDEP*). Model (6) controls for domestic institutional investors' ownership (*INS_OWN*). The regression result of a sub-sample excluding

companies in the manufacturing sector is displayed in Model (7). All variables are defined in Appendix A. Industry and year dummies are included in each regression. We cluster standard errors by year and firm. The 0.01, 0.05, and 0.1 significance levels are denoted by ***, **, and * (two-tailed), respectively.

Dep. Var. = ICI	Dynamic panel-data estimation
	System GMM
	(1)
QFII_DUMMY	0.5563***
	(3.5111)
LAG_ICI	0.2550***
	(51.4778)
FIRMSIZE	2.6048***
	(21.8397)
FIRMAGE	-5.1031***
	(-25.3502)
LEVERAGE	-10.4796***
CD OW/TH	(-16.3436)
GROWTH	-0.5961***
LOCC	(-8.0/86)
1055	-12.0942^{+++}
OWN CON	(-02.9737) 4.6107***
own_con	(-12,6155)
SOE	5 5954***
552	(14.2619)
BOARDSIZE	6.2281***
	(7.0032)
INDEP	3.7743*
	(1.7356)
DUALITY	-0.7758***
	(-3.6860)
MEETING	0.2989
	(1.5865)
BIG4	-2.0637***
	(-3.0742)
_constant	-26.2760***
	(-4.8469)
Sargan test over-identification (p-value)	0.000
Difference-in-mansen test of exogeneity (p-value)	0.151 78402.07
	20,732
No. of Firms	20,732
10.01111115	2,JJ+

 Table 7 Dynamic GMM approach

Notes: This table displays the result from the dynamic GMM approach. We treat *QFII_DUMMY*, *LAG_ICI*, *FIRMSIZE*, *FIRMAGE*, *LEVERAGE*, *GROWTH*, *LOSS*, *OWN_CON*, *SOE*, *BOARDSIZE*, *INDEP*, *DUALITY*, *MEETING*, and *BIG4* as endogenous variables. Levels of these variables, which are lagged twice, are used as instruments in the first-differenced equation, and first-differences of these same variables that are lagged once, as additional instruments in the level equation. Z-statistics are displayed in parentheses. All variables are defined in Appendix A. The 0.01, 0.05, and 0.1 significance levels are denoted by ***, **, and * (two-tailed), respectively.

Table 8 PSM approach

Panel A: Estimated	ATT							
	ATT (T-stat)	Treatment group	Control group	No. of Obs.	Treatment group: Control group			
Dep. Var. = ICI		(mean)	(mean)					
QFII_DUMMY	1.1196**(2.39)	69.3921	68.2725	19,610	1,765 : 17,845			
Panel B: Univariate balanced test for pairs of treatment and control groups after matching								
Matching criteria	Treatment group	Control group	%bias reduction	Difference	p-Value			
	(mean)	(mean)			-			
FIRMSIZE	22.3130	22.3230	98.6	-0.0100	0.844			
FIRMAGE	2.1682	2.1987	29.3	-0.0305	0.159			
LEVERAGE	0.4584	0.4506	67.2	0.0078	0.258			
GROWTH	0.2100	0.1913	-372.4	0.0187	0.262			
LOSS	0.0669	0.0771	80.1	-0.0102	0.241			
OWN_CON	0.2024	0.2057	91.1	-0.0033	0.621			
SOE	0.6147	0.6147	100.0	0.0000	1.000			
BOARDSIZE	2.3175	2.3169	98.4	0.0006	0.926			
INDEP	0.3655	0.3635	-862.4	0.0020	0.254			
DUALITY	0.1892	0.1847	74.2	0.0045	0.730			
MEETING	2.2566	2.2611	69.4	-0.0045	0.693			
BIG4	0.1337	0.1331	99.4	0.0006	0.961			

Panel C: PSM sample

Dep. Var. =	ICI
	(1)
QFII_DUMMY	1.3738***
	(2.9305)
FIRMSIZE	4.5563***
	(6.7914)
FIRMAGE	-0.3690
	(-0.5715)
LEVERAGE	-11.9889**
	(-2.5394)
GROWTH	2.5203***
1.000	(2.6545)
L033	-12.0///*****
OWN CON	(-3.5313)
Own_con	(1,1700)
SOF	(1.1700) 2 272/**
SOE	(-2.1596)
BOARDSIZE	5 0006
DOTINDOLLE	(1.1841)
INDEP	-2.6073
	(-0.3648)
DUALITY	-0.8456
	(-0.7496)
MEETING	-1.2092
	(-0.8621)
BIG4	1.7716
	(1.5225)
_constant	-19.9102
	(-1.0247)
Year-FE	Yes
Industry-FE	Yes
NO. 01 ODS.	19,610
Adj. K-square	0.452

Notes: Panel A reports the estimated ATT, which is the difference between the average internal control index score of treatment companies and that of the control companies. Panel B presents the univariate balanced test for pairs of treatment and control companies after matching. All matching criteria are all in year *t-1*. Panel C reports the result by re-estimating Eq. (1) based on the PSM sample. All variables are defined in Appendix A. We cluster standard errors by firm and by year. The 0.01, 0.05, and 0.1 significance levels are denoted by ***, **, and * (two-tailed), respectively.

Dep. Var. =	ICI	ADJ_ICI	NO_DEFI
	N	/oice of QFII-licensed investors	
	(1)	(2)	(3)
QFII_TOPTEN	0.6643**	0.8021***	-1.0821**
	(2.3764)	(2.8663)	(-2.2685)
QFII_NON_TOPTEN	-0.0962	-0.0811	-0.9513**
	(-0.1642)	(-0.1385)	(-1.9735)
FIRMSIZE	4.6360***	4.3430***	-0.0088
	(43.6163)	(40.6417)	(-0.0741)
FIRMAGE	-1.6023***	-1.4512***	-0.1752
	(-10.7686)	(-9.8082)	(-1.0910)
LEVERAGE	-15.6711***	-14.9714***	0.4958
	(-26.5275)	(-25.1032)	(0.9706)
GROWTH	-0.4729**	-0.4897**	-0.1094
	(-2.0834)	(-2.1606)	(-0.8574)
LOSS	-16.0219***	-15.3152***	-0.3537
	(-38.2151)	(-36.3391)	(-1.3449)
OWN_CON	2.8322***	3.0861***	-0.5395
	(5.0672)	(5.4941)	(-0.7007)
SOE	-0.2121	-0.3115	0.9321**
	(-1.0611)	(-1.5540)	(2.4535)
BOARDSIZE	0.6049	0.3935	2.2907***
	(1.1343)	(0.7327)	(2.8817)
INDEP	1.7789	1.3481	0.8593
	(1.0358)	(0.7803)	(0.5231)
DUALITY	0.0569	0.0696	0.1863
	(0.2710)	(0.3320)	(0.5724)
MEETING	-0.2740	-0.0583	2.4233
	(-1.0491)	(-0.2221)	(1.6202)
BIG4	0.1584	0.1515	6.4180**
	(0.4436)	(0.4289)	(2.4334)
_constant	-21.5078***	-80.2325***	-11.8938***
	(-8.8364)	(-32.6511)	(-2.9274)
Year-FE	Yes	Yes	Yes
Industry-FE	Yes	Yes	Yes
No. of obs.	22,310	22,310	20,145
Adj. R-square	0.380	0.345	0.010

Notes: All variables are defined in Appendix A. We cluster standard errors by firm and year. All regressions control for industry and year fixed effects. The 0.01, 0.05, and 0.1 significance levels are denoted by ***, **, and * (two-tailed), respectively.

Dep. Var. =	ROA	ROA	
	(1)	(2)	
QFII_DUMMY	0.0148***	-0.0059	
	(11.0853)	(-0.4683)	
ICI	0.0028***	0.0027***	
	(3.8187)	(3.4505)	
QFII×ICI		0.0032*	
		(1.6515)	
FIRMSIZE	0.0043***	0.0043***	
	(6.9490)	(6.9616)	
LEVERAGE	-0.0351***	-0.0352***	
	(-9.0655)	(-9.0972)	
OCF	0.2130***	0.2131***	
	(33.9653)	(33.9760)	
SOE	-0.0077***	-0.0077***	
	(-7.2371)	(-7.2186)	
GROWTH	0.0118***	0.0118***	
	(11.8039)	(11.8284)	
BIG4	0.0018	0.0018	
	(1.0263)	(0.9757)	
_constant	-0.1017***	-0.1007***	
	(-7.8560)	(-7.7285)	
Year-FE	Yes	Yes	
Industry-FE	Yes	Yes	
No. of obs.	22,281	22,281	
Adj. R-square	0.145	0.145	

Notes: This table displays the influence of the enhanced internal control quality attributable to QFIIs on operating performance, measured by *ROA*. To better interpret the magnitude of the internal control score, *ICI* is in the form of the natural logarithm. All variables are defined in Appendix A and lagged by one year. Industry and year dummies are included. We cluster standard errors by firm and year. All regressions control for industry and year fixed effects. The 0.01, 0.05, and 0.1 significance levels are denoted by ***, **, and * (two-tailed), respectively.

	ble definitions and data sources
--	----------------------------------

	DEFINITION
DEPENDENT VARIABLES	
ICI	Score of internal control ranging and all the initial scores are divided by 10. Source: Dibo Internal
	Control and Rick Management Database
	The industry making editor adjusted internal control index score is measured as the deduction of the internal
ADJ_ICI	The industry-median-adjusted internal control index scole is measured as the deduction of the internal
	control index score of a firm from the median score for all listed firms in the same industry for a given
	year.
NO_DEFI	The number of internal control deficiencies. Source: CSMAR.
RESTATEMENT	A dummy variable that equals 1 if the financial statements of a firm are restated in a given fiscal year,
	and 0 otherwise. Source: Annual reports of Chinese listed firms and Dibo Internal Control and Risk
	Management Database
NO DESTATEMENT	The number of financial restatements. Source: Annual reports of Chinese listed firms and Dibe
NO_KESIAIEMENI	The number of matchar restatements, Source, Annual reports of Chinese fisted fifths and Dioo
	Internal Control and Risk Management Database.
KEY INDEPENDENT VARIA	BLES
QFII_DUMMY	A categorical variable assigned a value of one if a Chinese listed company has QFII-licensed
	investment entities in a fiscal year; otherwise, it is set to zero. Source: Wind-Financial Platform.
OFII OWN	The percentage of outstanding shares held by OFII-licensed investment entities in a fiscal year.
	Natural logarithm of the total number of OFIL entities of a company in a fiscal year
	A astagorized usights at to one for companies with OEII licensed investment articles from a country.
QFII_III0IIW0I	A categorical variable set to one for companies with QFI-indexistent entities from a country
	with high institutional quality (with a WGI score equal to or higher than the median WGI score); this
	variable is assigned a value of zero for companies without QFII-licensed entities or companies with
	QFII-licensed investment entities from a country with low institutional quality (with a WGI score
	below the median WGI score) in a fiscal year. Source: Wind-Financial Platform, the State
	Administration of Foreign Exchange, and the WGI scores from the World Bank
	(https://info.worldbank.org/governance/wgi/Home/Documents)
OFU LOWWCI	A categorical variable assigned a value of one for companies with OEII licensed investment antitias
grn_Lowwon	A categorical variable assigned a value of the total companies with of the needling WCL access), this
	from a country with fow institutional quality (with a workscore below the median workscore); this
	variable is set to zero for companies without QFII-licensed investment entities or companies with
	QFIIs from a country with high institutional quality (with a WGI score equal to or greater than the
	median WGI score) in a fiscal year.
<i>QFII_HIGHWGI_OWN</i>	The sum of the percentage of outstanding shares held by QFII-licensed investment entities from a
~	country with high institutional quality (with a WGI score equal to or above the median level of sample
	countries in a fiscal year)
OFU LOWWCL OWN	The sum of the percentage of outstanding shares held by OFII licensed investment entities from a
QFII_LOW WOI_OWN	The sum of the percentage of outstanding shares here by QFIT-herensed investment entries from a
	country with low institutional quality (with a well index score below the median well).
QFII_TOPTEN	A categorical variable assigned a value of one if the percentage of QFII's shareholding is greater than
	or equal to that of the 10th shareholder of a company in a fiscal year; this variable is set to zero if the
	percentage of shareholding is less than that of the 10th shareholder or if a company does not have a
	QFII-licensed investment entity in the same year.
OFII NON TOPTEN	A categorical variable assigned a value of one if the percentage of OFII's shareholding is less than
2	that of the 10^{th} shareholder of a company in a fiscal year; this variable is set to zero if the percentage
	af OEU' a shareholding is greater than an again to that of the 10 th shareholder or if a company does not
	by a OEU licensed investment entity is the some year
CONTROL VARIABLES AND	have a QFT-incensed investment entity in the same year.
CONTROL VARIABLES AND	VARIABLES IN THE EXTENDED ANALYSIS
FIRMSIZE	Book value of total assets (in the form of natural logarithm).
FIRMAGE	The number of years since stock listing (in the form of natural logarithm).
LEVERAGE	Book value of total liabilities divided by the book value of total assets.
GROWTH	The percentage change in annual sales.
LOSS	A categorical variable assigned a value of one if the company in the prior year reported pegative pet
	income and it is set to zero otherwise
OWN CON	nucline, and it is set to be other wise.
01111_0011	ownership concentration is measured as the percentage of a company's outstanding shares that are
	owned by the largest shareholder.
SOE	A categorical variable that is set to one if a company's ultimate controlling shareholder is the central
	or a local government, or a government agency; otherwise, it is assigned a value of zero.
BOARDSIZE	Total number of directors in the boardroom, in the form of natural logarithm.
INDEP	The percentage of independent directors in the boardroom.
DUALITY	A categorical variable set to one if the CEO of a company and its chairnerson in the boardroom is the
	same and it receives a value of zero otherwise
MEETINC	Sume, and infectives a value of 200 only who.
	Natural logarithm of the humber of board meetings field each year.
B1G4	A categorical variable equal to one if the company is audited by a Big 4 auditor in a fiscal year;
	otherwise, it is set to zero. The Big Four accounting organisations include PricewaterhouseCoopers,
	Ernst & Young, Deloitte, and KPMG.
AUDIT_SIZE	Natural logarithm of the total number of audit committee members.
AUDIT INDEP	The percentage of independent members on the audit committee.
INS OWN	The percentage of outstanding shares held by domestic institutional investors
ROA	Farming before interacts and taxas (FBIT) divided by the book value of total assate
ACA	Consider a set form set to be beyong factor of the form
UCF	Operating cash now over the book value of total assets.

App	endix B.	Additional	information on	n sample distribution
-----	----------	------------	----------------	-----------------------

	No. of Obs. with the pro-	esence of QFIIs	
	Institutional	quality	
Year	High WGI countries	Low WGI countries	Total
2005	6	7	13
2006	31	43	74
2007	86	72	158
2008	63	64	127
2009	39	67	106
2010	116	63	179
2011	47	131	178
2012	71	52	123
2013	69	73	142
2014	97	98	195
2015	124	100	224
2016	87	82	169
2017	134	134	268
Total	970	986	1.956

Notes: We obtain the country-level WGI scores from the World Bank and calculate the median value for each year. If a QFII is from a country with a WGI score equal to or above the median value in a fiscal year, it is then classified as high institutional quality; otherwise, it is sorted into low institutional quality group.

Dep. Var. =	QFII_DUMMY
	(1)
	Probit regression
LAG_ICI	0.0024
	(1.5000)
QFII_DUMMY	1.1300***
	(26.5950)
FIRMSIZE	0.1480^{***}
	(8.6922)
FIRMAGE	0.0255
	(0.9875)
LEVERAGE	-0.5325***
CDOWTH	(-5./1/5)
GROWTH	0.1100^{+++}
1000	(4.3977)
L035	-0.0939
OWN CON	(-1.3007) 0.3315***
	(3 ///0)
SOE	0.0757**
501	(2,0091)
BOARDSIZE	0.0828
	(0.8544)
INDEP	-0.0275
	(-0.0943)
DUALITY	0.0799**
	(2.1080)
MEETING	-0.0263
	(-0.5924)
BIG4	0.1091*
	(1.7234)
_constant	-4.7663***
	(-12.5863)
Year-FE	Yes
Industry-FE	Yes
No. of obs.	19,586
Pseudo K-square	0.152
waid Cm-Square	1412.39

Appendix C. Additional test on the reverse causancy iss	Un the reverse causality issue
---	--------------------------------

Notes: This table reports the result of a probit regression of the influence of lagged internal control quality on the likelihood of the presence of a QFII in the subsequent year. The key independent variable is the overall firm-level internal control index score (*LAG_ICI*). All variables are defined in Appendix A. Industry and year dummies are included in each regression. The 0.01, 0.05, and 0.1 significance levels are denoted by ***, **, and * (two-tailed), respectively.